Author's response to reviews

Title: Essential Infantile Esotropia: Postoperative motor outcomes and Inferential Analysis of strabismus surgery

Authors:

Adriano Magli (magli@unina.it)
Roberta Carelli (robertacarelli@hotmail.it)
Francesco Matarazzo (francesco.matarazzo@gmail.com)
Dario Bruzzese (dario.bruzzese@unina.it)

Version: 2 Date: 15 July 2013

Author's response to reviews:

Dear Editor,

these are the revisions made by the authors. The points 1 and 2 are in "Methods". Points 3-4-5 follows the discussion.

Kind regards

(1) Ethics.
For this retrospective study no ethics approval was necessary, all surgical procedures were conducted in accordance with Italian and European regulations.

(2)
No access to databases was necessary for this study.

(3) Authors' contributions
A.M. conceived and designed the study, together with R.C. Furthermore, A.M. critically revised the manuscript for intellectual content and gave final approval of the version to be published.
R.C. carried out the interpretation of the data and drafted the manuscript.
F.M. participated in the design of the study, provided the acquisition of data and cooperated to the manuscript drafting.
D.B. participated in the design of the study and performed the statistical analysis of the examined data.
All authors read and approved the final manuscript.

(4) Competing interests
The authors of this manuscript have no financial competing interests. They do not have any other non-financial interests that influence the results and
discussion of this paper.

(5) Acknowledgements
No Acknowledgements